Author's response to reviews

Title: Audit of therapeutic interventions in inpatient children using two scores: are they evidence-based in developing countries?

Authors:

Nilton Y Carreazo (yhuro@hotmail.com)
Carlos A Bada (ifocar@yahoo.com)
Juan P Chalco (jpcho33@yahoo.com)
Luis Huicho (lhuicho@viabcp.com)

Version: 5 Date: 16 December 2004

Author's response to reviews:

Lima, December 16, 2004

Dr. Alison Z Shaw, Ph.D.
Assistant Editor
editorial@biomedcentral.com

Dear Dr. Shaw,

Many thanks for the further format reviews to our manuscript.

You requested that where all authors have the same affiliation, the superscript numbers from the author list should be removed. This has been done now.

For Tables, you requested to convert all tables in portrait format. This is done now.

We are re-submitting the revised manuscript.

Yours sincerely,

Luis Huicho, MD